Massachusetts Institute of Technology Handout 29 6.857: Network and Computer Security November 19, 1998 Professor Ronald L. Rivest

Readings for Lecture 20

Attached are the following readings:

• readings from AUSCRYPT92, pages 244–251 [2]

• readings from Computer Security Reference Book, pages 37–73 [1]

References

[1] Computer Security Reference Book. Butterworth-Heinemann Ltd., 1992.

[2] Atsushi Fujioka, Tatsuaki Okamoto, and Kazuo Ohta. A practical secret voting scheme for large scale elections. In J. Seberry and J. Pieprzyk, editors, Advances in Cryptology– AUSCRYPT ’92, volume 718 of Lecture Notes in Computer Science, pages 244–251. Springer-Verlag, 1992.